Author's response to reviews

Title: The Potential Biomarkers in Predicting Pathologic Response of Breast Cancer to Three Different Chemotherapy Regimens

Authors:

Linbo Wang (drlinbowang@hotmail.com)
Zhinong Jiang (jzn1008@163.com)
Meihua Sui (suim@musc.edu)
Jianguo Shen (shenjianguo76623@hotmail.com)
Chaoyang Xu (xuchaoyang2001@126.com)
Weimin Fan (fanw@musc.edu)

Version: 3 Date: 30 June 2009

Author's response to reviews: see over
Dear Dr. Andrea Bucceri:

Thank you very much for your kind letter about our manuscript (MS: 1179592495246112). According to your suggestion, we have specified the informed consent and ethical approval of this study in the Methods section of the revised manuscript (highlighted in red).

Thank you for your time and consideration. Please do not hesitate to let us know if you have any questions with this revised manuscript.

Sincerely,

Linbo Wang, M.D., Ph.D.  
Professor  
Department of Surgical Oncology  
Sir Run Run Shaw Hospital  
Zhejiang University College of Medicine

Weimin Fan, M.D., M.P.H.  
Professor  
Pathology and Laboratory Medicine  
Medical University of South Carolina

June 30, 2009